Author's response to reviews

Title: Does decreasing serum uric acid level prevent hypertension? --A nested randomised controlled trial in cohort study: Rationale, methods and baseline characteristics of study cohort

Authors:

Kun Song (songkun_2012@126.com)
Yuan Wang (gw_928@163.com)
Wenli Lu (luwenli1234@163.com)
Qing Zhang (zhangqing_201206@163.com)
Wen Li (liwen828@163.com)
Huanli Jiao (jiaohuanli2012@163.com)
Guowei Huang (huangguowei@tmu.edu.cn)
Guolin Wang (wang_guolin@hotmail.com)

Version: 2 Date: 3 January 2013

Author's response to reviews:

1. We added the trial registration number after the abstract.
2. We stated we have obtained informed written consent from the study participants.